Author's response to reviews

Title: Acute renal failure as a form of presentation of sarcoidosis at the young adult: a case report

Authors:

Joana Rema (joana.rema@gmail.com)
Margarida Carvalho (amcocs@gmail.com)
Raquel Vaz (mraquelvaz@gmail.com)
Margarida Fonseca (magutas@gmail.com)
Susana Sampaio (susana.sampaio@sapo.pt)
Augusta Praça (augustapraca@gmail.com)
Margarida Freitas-Silva (m.corisca@gmail.com)
Teresa Cardoso (mtcpteresa@gmail.com)

Version: 5 Date: 12 June 2014

Author's response to reviews: see over
Author's response to reviews

Title: Acute renal failure as a form of presentation of sarcoidosis at the young adult: a case report

Authors:
Joana Rema (joana.rema@gmail.com)
Margarida Carvalho (amcocs@gmail.com)
Raquel Vaz (mraquelvaz@gmail.com)
Margarida Fonseca (magutas@gmail.com)
Susana Sampaio (susana.sampaio@sapo.pt)
Augusta Praça (augustapraca@gmail.com)
Margarida Freitas-Silva (m.corisca@gmail.com)
M Teresa Cardoso (mtcpteresa@gmail.com)

Version: 4 Date: 12 June 2014

Author's response to reviews: see over
12 June, 2014.

The Biomed Central Editorial Team

Object: MS: 7088330191065311- Acute renal failure as a form of presentation of sarcoidosis at the young adult: a case report. Dr Joana Rema et al.

Thank you for consideration of our manuscript for publication in your journal.

We have reviewed the above manuscript according to your reviewer’s comments.

Reviewer 1 (Dr Guillaume Armengol)

1. “Just please pay attention to the numbers: delete the comma and place a dot (100.000 and not 100,00. 5.6 and not 5,6 etc...). (PTH) is not correctly placed into the text and must be placed just after the word parathormone (without capital P by the way)”

Done

Reviewer 2 (Dr E. Papadavid) No further changes suggested.